Author's response to reviews

Title: Asian-Specific Total Knee System: 5-14 year Follow-up Study

Authors:

Kunihiro Hosaka (kunihiro619@hotmail.com)
Shu Saito (saitou.syu@nihon-u.ac.jp)
Takao Ishii (ishii.takao@nihon-u.ac.jp)
Sei Mori (moriseimori@gmail.com)
Takanobu Sumino (sumino.takanobu@nihon-u.ac.jp)
Yasuaki Tokuhashi (tokuhashi.yasuaki@nihon-u.ac.jp)

Version: 5 Date: 30 October 2011

Author's response to reviews: see over
Dear Editor:

Thank you for accepting and publishing our manuscript. I appreciate you very much. Please find our point-by-point response to your comment below.

Major revisions (we require the author to make these changes)

Authors' contributions: Please indicate all authors have read and approved the final manuscript.

Response: We added “All authors have read and approved the final manuscript.” in Authors' contributions.

Tables: tables must be created using a Tables tool, and should not be linked files, figures, or generated with tabbed text. Tables must be divided into rows and columns. Please ensure that text alignment in columns is clear and consistent. We recommend the MS Word auto-format template: simple 1. Please do not include vertical lines in your tables.

Response: We changed Table 1-3 using a Tables tool and MS Word. Please show our Table.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Author list: Please remove titles.

Response: We removed titles.

Line numbers: Please remove these.

Response: We removed line numbers.

Highlighting/tracking: Please remove from the manuscript.

Response: We removed Highlighting/tracking parts.
Structure: Change material and methods to methods.
Response: We changed material and methods to methods

Figures: Please crop figures as close as possible so there is no white space around the figures.
Response: I am sorry, but we could not crop figures as close as possible so there is no white space around the figures.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Response: We checked some typographical errors in our manuscript.